Author's response to reviews

Title: The effect of ursodeoxycholic acid in Liver functional restoration of patients with obstructive jaundice after endoscopic treatment: a prospective, randomized, and controlled study

Authors:

Enver H Fekaj (enverfekaj@hotmail.com)
Arben Gjata (arbengjata@msn.com)
Mehmet Maxhuni (mehmetmaxhuni@hotmail.com)

Version: 5 Date: 19 July 2013

Author's response to reviews: see over
Dear Editors,

**Our response to the Editorial concerns:**

1) Each manuscript author will send you an email, confirming that they agree to the changes of the study protocol, although, the two authors of this manuscript (Arben Gjata, and Mehmet Maxhuni) have sent to you their agreements to the changes that have been made to the version of the manuscript submitted 14th July 2013.

2) In this version of our study protocol have not been made any changes in response to the comments of reviewer 2, because the reviewer 2 does not require any changes.

3) We have designed the manuscript conforms to the journal style.

Best wishes,

Enver Fekaj